Author’s response to reviews

Title: Infarcted Warthin tumor with mucoepidermoid carcinoma-like metaplasia: a case report and literature review

Authors:
Kenji Yorita (kenjiyorita@gmail.com)
Hideyuki Nakagawa (hktokunaka1119@yahoo.co.jp)
Katsushi Miyazaki (katsushi@bronze.ocn.ne.jp)
Junya Fukuda (fukudaj5@gmail.com)
Satoshi Ito (s-itou@kfy.biglobe.ne.jp)
Makoto Kosai (mkosai@tenor.ocn.ne.jp)

Version: 1 Date: 22 Nov 2018

Author’s response to reviews:

Referee: 1

Reviewer reports:

Reviewer #1: This is an excellent case report with good review. The authors are applauded for their work. We'd like to see more radiographs, perhaps bone window of the CT images. Only soft tissue window was provided.

Thank you.

Our response: Thank you for your suggestion. We have added a CT image in bone window to Figure 1. The bone tissue appeared not to have some diseases.

Reviewer #2: This manuscript lets us know the importance of genetic studies for differentiating different kinds of tumors which should be noted in clinical practice.

Thank you for your comment.